Author's response to reviews

Title: Comparison of early-, late-, and non-participants in a school-based asthma management program for urban high school students

Authors:

Christine LM Joseph (cjoseph1@hfhs.org)
Jacquelyn Saltzgaber (jsaltz1@hfhs.org)
Suzanne L Havstad (shavsta1@hfhs.org)
Christine C Johnson (cjohnso1@hfhs.org)
Dayna Johnson (djohnso3@hfhs.org)
Edward L Peterson (epeters1@hfhs.org)
Gwen Alexander (galexa2@hfhs.org)
Michael Couper (mcouper@umich.edu)
Dennis R Ownby (downby@mail.mcg.edu)

Version: 5 Date: 9 May 2011

Author's response to reviews: see over
May 9, 2011

Doug Altman, Editor-in-Chief
Trials
BioMed Central Ltd
Floor 6, 236 Gray's Inn Road
London WC1X 8HB
United Kingdom

RE: Trials, MS: 2089864254943765, Formatting changes required: Comparison of early-, late-, and non-participants in a school-based asthma management program for urban high school students

Dear Dr. Altman,

Thank you again for the opportunity to publish in Trials. Our corrected resubmission includes:

1. Author list: the list of authors in the manuscript appears exactly as they are in the submission system, both in style and order.

2. Tables: we have removed the shading from the tables.

3. References: We have cited references in text using consecutive numbers in square brackets and do not use superscript numbers. References appear before the full stop.

4. Footnotes: We incorporated the footnotes into the text of the manuscript.

Please do not hesitate to contact me if there are additional questions about our manuscript. Thank you.

Sincerely,

Christine LM Joseph, PhD
Epidemiologist
On behalf of authors